XIV CONVEGNO ANNUALE DELL'ASSOCIAZIONE ITALIANA DEI PROFESSORI UNIVERSITARI DI DIRITTO COMMERCIALE "ORIZZONTI DEL DIRITTO COMMERCIALE"

"Imprese, mercati e sostenibilità: nuove sfide per il diritto COMMERCIALE" Roma, 26-27 maggio 2023

GAIA BALP

Professoressa associata di diritto commerciale, Università Bocconi Giovanni Strampelli

Professore ordinario di diritto commerciale, Università Bocconi

Institutional Investor ESG Engagement: The European Experience

Summary

1.	Introduction
2.	The Call for Investor Engagement, and the Role of Stewardship and Corporate Governance Codes in Underpinning Board-Shareholder Engagement
3.	The SHRD II as the Regulatory Milestone for Shareholder Engagement Scaling Down Potential Regulatory Hurdles to Board-Shareholder Dialogue 7
4.	Sustainable Investing as a Driver for Institutional Investor Engagement: the SFDR and the Taxonomy Regulation
5.	The Issuer Side of Sustainability: from the NFRD to the CSRD, through to the CSDD
6.	Uncertainties Surrounding Institutional Investors' Walk to Sustainable Finance
	A. Resource and Cost Constraints to Active Stewardship, and the Problem of Collective Action
	B. Dependency on End-Investor Preferences and Priorities
	C. External Limitations to Institutional Investors' ESG Investing and Stewardship
	D. ESG Voting in Practice
7.	Conclusions

1 Introduction

The view according to which institutional investors are key actors in corporate governance of publicly listed firms is long since established,¹ and increasingly topical. As they hold relevant stakes in an incalculable number of publicly listed firms globally, and, collectively, they are often a corporation's major shareholders,² leading institutional investors are very often called upon to perform an effective stewardship function and exert influence over investee companies.³ In line with the fiduciary duties they owe to end-clients (including the duty to make informed voting decisions) and the objective to improve investment returns and prevail over competitors, asset managers' engagement is primarily aimed at preserving and enhancing the overall value of the firms they are invested in and, by doing so, at maximizing shareholder value for their clients, whose money they ultimately manage.

Over the last several years, institutions' appetite for engagement has grown. Not only do asset managers and asset owners – especially larger ones – regularly vote portfolio shares at shareholders' meetings. Institutions' active ownership playbook also largely relies on meeting the board, or management, of portfolio companies to discuss key issues, convey their concerns and views and urge the board to collaboratively respond to investor demands. Amongst other tools (see Article 3g(1)(a) SRD II⁴), private dialogue with corporate directors that takes place 'behind the scenes' is an instrument of investor engagement that has gained much traction in recent years.⁵

Indeed, available evidence shows that private board-shareholder dialogue has grown into one of the levers of engagement most popular among institutional shareholders.⁶ As further studies suggest, privately meeting company directors and management (and, more generally, activism behind the scenes) can actually create value and have an impact on issuers' decisions.⁷

_

¹ See, e.g., European Commission (2012), p 8.

² See Griffin (2020), p 411.

³ European Fund and Asset Management Association (2018), p.3.

⁴ Directive (EU) 2017/828 of the European Parliament and of the Council of 17 May 2017 amending Directive 2007/36/EC as regards the encouragement of long-term shareholder engagement, OJ L 132, 20.5.2017, pp 1-25 (so-called Shareholders' Rights Directive II – SRD II).

⁵ See European Securities and Markets Authority (2019), p 55.

⁶ Ibid.; McCahery et al. (2016), pp 2911-2912.

⁷ See Becht et al. (2010), pp 3108-3118; Carleton et al. (1998), p 1335; Bonacchi et al. (2022).

Seeking to increase the amount and value of the assets under management to earn enhanced financial returns remains the primary and fundamental objective for any institutional investor. However, the focus of institutions' investment and engagement strategies has changed over time to include broader, non-financial objectives as well. Different factors contributed to bringing about this shift in institutions' approach to investment management in regard to portfolio companies.

First, the theory of shareholder value maximization, and shareholder primacy, according to which corporations are bound to primarily serve the interests of their shareholders,⁸ has attracted increased criticism. Also due to the insufficiency of government regulation to correct market failures and advance public welfare⁹, calls on corporations to pursue a broader purpose and act according to societal interests to generate long-term value for all stakeholders have multiplied.¹⁰ In particular, companies are insistently being urged to play an active role in the global transition towards a sustainable economy and, especially, in contributing to mitigating the consequences of climate change.

Second, beyond controlling stockholders, ownership at publicly listed companies is becoming increasingly concentrated in the EU, owing to the unstoppable growth in intermediated investments and the concentration of the asset management industry itself. According to EFAMA, at the end of 2020, 'European asset managers held 25.8% of the debt securities and 26.9% of the listed shares issued by European residents', and 'the industry holding of listed shares represented 34.5% of the free-float market capitalisation in Europe'. 11 Moreover, despite the fact that evidence about European markets in regard to common ownership is rare, a study promoted by the European Commission found that, over the period from 2007 to 2017, '[t]he number of listed firms that are cross-held by block-holders has been increasing from about 15.5 thousand to about 17.5 thousand', and their proportion over the whole set of firms is rather stable on 'values around 67%' - such results being 'very close to US-based indicators'. 12 The top-ranking investors, presenting maximum values in the density-based indices, were found to be 'Bank of New York Mellon, BlackRock, Fidelity, JP Morgan Chase, Norway, State Street and Vanguard', indicating that mainly US-based large investors 'show a clear dominance over the set of listed firms active in Europe'. 13

As assets, the asset management industry and corporate ownership tend towards concentration, institutions' stewardship generally has turned

⁸ See, famously, Friedman (1970).

⁹ See Lund (2022), pp 7, 25.

¹⁰ Business Roundtable (2019). See also World Economic Forum (2019).

¹¹ European Fund and Asset Management Association (2021), p 26.

¹² Rosati et al. (2020), p 52.

¹³ Ibid., p 57.

inescapable. Not surprisingly, regulatory action has reckoned with the increases in institutional ownership and ownership concentration by enhancing institutions' 'responsibilities' in performing the corporate governance functions associated with share ownership, including the expectation that leading investors, as corporations' main shareholders beyond controlling stockholders, take a proactive part in driving a shift towards corporate sustainability and favoring the transition to a net-zero economy. As explicitly acknowledged by the European Commission, the scale of the investment required to meaningfully address climate change and environmental degradation and avoid major adverse implications for financial stability 'is well beyond the capacity of the public sector'. Hence, 'the alignment of all sources of finance – public and private, national and multilateral – is required'. 15

As a consequence of such evolutionary patterns, the understanding of the role of institutional investors as corporate stewards has changed. Only a few years ago, stewardship focused essentially, if not exclusively, on portfolio firms' corporate governance. Today, stewardship and investment strategies adopted by many institutional investors embrace environmental and social standards as well, along governance standards (ESG), as a means to incorporate the risk assessment of long-term environmental, social and governance challenges and developments, and foster investee companies' more responsible conduct to positively impact the environment and society at large. With a view to capturing and attracting ever-widening cohorts of end-clients more sensitive to environmental and social concerns and to increasing assets under management, major institutional investors now regularly uphold that ESG factors are at the core of their engagement policies and practice, and claim they are indeed willing to play a major part in pursuing sustainability and, particularly, in fighting climate change.

On the normative side, enhanced awareness of the systemic challenges posed by environmental and social issues, addressing many of which can be delayed no further, has driven regulatory action undertaken at the EU level more strongly by far than in any other jurisdiction. Institutional investors are viewed as pivotal in achieving the so-called European Green Deal, by which the European Commission intends to make Europe 'a modern, resource-efficient and competitive economy', put it 'on a new path of sustainable and inclusive growth' and ultimately make it 'the first climate-neutral continent'.¹⁷ According to the Commission, '[t]he private sector will be key to financing the green transition. Long-term signals are

¹⁴ Strine, Jr (2020).

¹⁵ European Commission (2021a).

¹⁶ See Boffo and Patalano (2020), p 14.

¹⁷ See European Commission (2021a).

needed to direct financial and capital flows to green investment and to avoid stranded assets'.¹⁸

Against this backdrop, the question as to whether institutional investors really can effectively play the role of the engagement and sustainability champions EU-level legislation has tailored for them remains however controversial. This article examines whether the EU rules and standards are enough to actually encourage institutions' conduct to that effect, and whether institutions are well equipped to play such role.

This article proceeds as follows. Part 2 introduces the concepts of shareholder engagement and stewardship and their actual relevance in the practice of shareholder-issuer relationships. Part 3 turns to analyzing the relevant EU regulatory framework which underpins private dialogue as a tool to promote institutional investors' active share ownership. Part 4 focuses on sustainable investing as a potentially powerful driver of institutions' engagement, and the new disclosure obligations imposed on them by the Sustainable Finance Disclosure Regulation (SFDR)¹⁹ and the Taxonomy Regulation.²⁰ Issuer disclosures under the Non-Financial Disclosure Directive (NFRD21 and the Commission's proposals for a Corporate Sustainability Disclosure Directive (CSRD)²² and for a Corporate Sustainability Due Diligence Directive (CSDD)²³ are accounted for in Part 5, mainly from the standpoint of institutional investors as the primary users of corporate sustainability information. Part 6 analyses the incentives structure which is typical of (different kinds of) institutional investors as a foundation based on which to evaluate, by also taking account of a number of further factors, largely independent of the asset managers themselves,

¹⁹ Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector, OJ L 317, 9.12.2019, pp 1-16.

¹⁸ Ibid., pp 16-17.

 $^{^{20}}$ Regulation (EU) 2020/852 of the European Parliament and of the Council of 18 June 2020 on the establishment of a framework to facilitate sustainable investment, and amending Regulation (EU) 2019/2088, OJ L 198, 22.6.2020, pp 13-43.

²¹ Directive 2014/95/EU of the European Parliament and of the Council of 22 October 2014 amending Directive 2013/34/EU as regards disclosure of non-financial and diversity information by certain large undertakings and groups, OJ L 330, 15.11.2014, pp 1-9.

²² European Commission (2021b). The CSRD was adopted by the European Parliament on 10 November 2022, and the Council is expected to adopt the proposal on 28 November 2022, after which it will be signed and published in the EU Official Journal. See Position of the European Parliament adopted at first reading on 10 November 2022 with a view to the adoption of Directive (EU) 2022/... of the European Parliament and of the Council amending Regulation (EU) No 537/2014, Directive 2004/109/EC, Directive 2006/43/EC and Directive 2013/34/EU, as regards corporate sustainability reporting (P9_TC1-COD(2021)0104), https://www.europarl.europa.eu/doceo/document/TA-9-2022-0380_EN.html#title2.

²³ European Commission (2022a).

institutions' actual readiness, and ability, to effectively take on the stewardship role they are assigned by the EU. Part 7 concludes.

2 The Call for Investor Engagement, and the Role of Stewardship and Corporate Governance Codes in Underpinning Board-Shareholder Engagement

Before being first explicitly embraced by EU legislation, the concepts of shareholder engagement and, as a broader category, stewardship grew popular in the EU with the rise of best practice codes and principles for institutional investors. In effect, more active institutional share ownership was, and still is, largely underpinned by soft regulation in the form of both stewardship and corporate governance codes targeting institutional investors and publicly listed corporations.²⁴ Indeed, all main stewardship codes, and many corporate governance codes as well, look at director-shareholder dialogue as an important lever for engagement.

While several methods of engagement exist,²⁵ interaction with corporate directors and managers behind closed doors is key to investors.²⁶ More confrontational strategies, such as submitting resolutions at general meetings or voting against resolutions proposed by the management, are generally considered only by way of escalation, if and when portfolio companies fail to constructively respond to shareholders' demands and concerns, as is also generally recommended by stewardship principles.²⁷

For example, according to the UK Stewardship Code 2020, signatories are expected to engage with issuers to maintain or enhance the value of the assets under management, and engagement methods include meeting the chair or other board members, as well as the company's management.²⁸ The EFAMA Code includes a similar recommendation.²⁹ Institutional investors should interact with investee companies on an ongoing basis to protect and secure value over the long term; discussions may entail meeting with the chief executive officer, senior independent directors, or the chair of the supervisory board, as the case may be, or with other independent directors or board members. More generally, when investors are concerned about the company's strategy and performance, the EFAMA Code recommends that they should seek to ensure that the appropriate members of an investee company's board are made aware of them.

²⁵ See Financial Reporting Council (2019), p 17.

²⁴ See Hopt (2011), pp 10-14.

²⁶ European Securities and Markets Authority (2019).

²⁷ European Fund and Asset Management Association (2018) p 7.

²⁸ Financial Reporting Council (2019), p 17.

²⁹ European Fund and Asset Management Association (2018), pp 6-8.

On the issuer side, corporate governance codes do typically include principles and recommendations concerning director-shareholder dialogue that run parallel to those by which stewardship codes target institutional investors as shareholders. The UK Corporate Governance Code recommends that

[i]n addition to formal general meetings, the chair should seek regular engagement with major shareholders in order to understand their views on governance and performance against the strategy. Committee chairs should seek engagement with shareholders on significant matters related to their areas of responsibility. The chair should ensure that the board as a whole has a clear understanding of the views of shareholders.³⁰

Provisions similar to those comprised in the UK Corporate Governance Code can be found in corporate governance codes adopted in other Member States as well. The 2020 Belgian Code on Corporate Governance recommends that the board 'ensure an effective dialogue with shareholders and potential shareholders through appropriate investor relation programmes, in order to achieve a better understanding of their objectives and concerns'31 and 'encourage shareholders, and in particular, institutional investors, to communicate their evaluation of the company's corporate governance prior to the general shareholders' meetings and at least through participation in the general shareholders' meeting'.32 One further example is that of the Italian Corporate Governance Code, under which '[t]he board of directors promotes dialogue with shareholders and other stakeholders which are relevant for the company, in the most appropriate way'.33 The Code also recommends that 'a policy for managing dialogue with the generality of shareholders, taking also into account the engagement policies adopted by institutional investors and asset managers' be adopted by the board of directors and be described in the yearly corporate governance report the board is required to file for public disclosure ahead of the shareholders' meeting.34

3 The SRD II as the Regulatory Milestone for Shareholder Engagement. Scaling Down Potential Regulatory Hurdles to Board-Shareholder Dialogue

³³ Italian Corporate Governance Committee (2020), Principle IV.

³⁰ Financial Reporting Council (2018), p 4.

³¹ Corporate Governance Committee (2019), p 24.

³² Ibid

³⁴ Ibid., recommendation No. 3 to Article 1.

Unlike in other jurisdictions such as the United States, the concept of shareholder engagement is explicitly encompassed in EU legislation since the 2017 SRD II. The European debate concerning engagement, including board-shareholder interaction, was actually initiated by the Commission's 2012 Action Plan on European company law and corporate governance,³⁵ which morphed into the proposal for a Directive reforming the Shareholders Rights Directive of 2007.³⁶ The first draft of the amending Directive already included provisions on director-institutional shareholder dialogue, which have been maintained in the final text of the SRD II.

In fact, Article 3g SRD II stipulates under a comply-or-explain rule that institutional investors and asset managers develop and publicly disclose an engagement policy describing how they integrate shareholder engagement into their investment strategy. Among other things, the engagement policy is to describe how institutions 'conduct dialogues with investee companies'.³⁷ Under the same rule, institutions are also required to publicly disclose, on an annual basis, how their engagement policy has been implemented, especially, but not limited to, as regards voting portfolio shares. Relatedly, Article 3j SRD II requires proxy advisors to yearly disclose, i.a., 'whether they have dialogues with the companies which are the object of their research, advice or voting recommendations and with the stakeholders of the company, and, if so, the extent and nature thereof'. The transparency obligations imposed on proxy advisors are intended to complement institutional investors' disclosures, based on the assumption that - institutions being proxy advisors' almost exclusive clients - service providers will engage with publicly listed firms by essentially taking the standpoint of client institutions, if not by acting on behalf of client institutions.

Hence, as regards the EU, the SRD II may be looked at as the regulatory milestone for engagement generally, and for board-shareholder dialogue particularly. As noted by the ESMA, the SRD II recognizes that engagement should be reinforced to reduce excessive managerial focus on short-term returns, and that encouraging long-term shareholder engagement is its key goal.³⁸

³⁵ European Commission (2012).

³⁶ European Commission (2014).

³⁷ One further reference to director-shareholder dialogue is included in Recital 49 to the SRD II, which recognizes that shareholders and investors can engage with the company on the implementation of the remuneration policy. Due to the introduction of 'say-on-pay' votes, executive compensation has become a privileged area of engagement both in the US and in Europe. See Thomas et al. (2012), p 1256.

³⁸ European Securities and Markets Authority (2019), p 53.

Against this backdrop, the preferences of institutional investors for board-shareholder dialogue as an effective lever for engaging with portfolio firms should not be cooled down by concerns associated with the application of further pieces of regulation which may, at first sight, seem to oppose that practice.³⁹

In fact, potential legal constraints to board-shareholder dialogue associated with the Market Abuse Regulation (MAR) 40 and, in particular, with issuers' disclosure obligation of non-public price-sensitive information and the ban on trading based on inside information, do not pose any absolute ban on board-shareholder dialogue. The same applies in regard to the principle of equal treatment of shareholders laid down by Article 17 of Directive 2004/109/EC (Transparency Directive) and by Article 46 of Directive 2012/30/EU.

First, regarding financial markets law, Recital 19 to the very MAR makes it clear that the mandatory disclosure and insider dealing regimes laid down by Articles 17 and 8 are 'not intended to prohibit discussions of a general nature regarding the business and market developments between shareholders and management concerning an issuer', given that '[s]uch relationships are essential for the efficient functioning of markets'. Articles 17 and 8 MAR only prohibit directors to disclose inside information within private meetings with shareholders, unless - under the selective disclosure exemption provided for by Article 10 MAR - shareholders agree to sign a confidentiality agreement and abstain from trading securities to which inside information relates. Importantly, however, during engagementrelated dialogue, investors generally seek to communicate their opinions and concerns regarding corporate governance and ESG matters, strategies or individual transactions, and are indeed unwilling to receive any inside information, since this would restrain their freedom to trade the securities concerned (Article 8 MAR).42

Second, board-shareholder dialogue, and the selective disclosure of (non-price-sensitive) information to participating investors are not

-

³⁹ See McCahery et al. (2016), pp 2920-2922; Fairfax (2013), pp 834-838. See also European Securities and Markets Authority (2019), p 64.

 $^{^{40}}$ Regulation 596/2014 of the European Parliament and of the Council of 16 April 2014 on market abuse (market abuse regulation) and repealing Directive 2003/6/EC of the European Parliament and of the Council and Commission Directives 2003/124/EC, 2003/125/EC and 2004/72/EC, [2014] OJ L173/1 (MAR).

⁴¹ Directive 2012/30/EU of the European Parliament and of the Council of 25 October 2012 on coordination of safeguards which, for the protection of the interests of members and others, are required by Member States of companies within the meaning of the second paragraph of Article 54 of the Treaty on the Functioning of the European Union, in respect of the formation of public limited liability companies and the maintenance and alteration of their capital, with a view to making such safeguards equivalent, OJ L315/74.

⁴² See, e.g., Soltes (2018), pp 148, 149.

incompatible with the principle of equal treatment of shareholders. Directors can lawfully consult with major shareholders, since the principle of equality applies to the shares, not to the shareholders as such. That view is consistent with the wording of Article 17 Transparency Directive, under which only shareholders 'who are in the same position' should be treated in an equal manner. Reasonably enough, as major shareholders cannot be considered to find themselves in the same position as retail investors holding negligible stakes, directors should be allowed to engage in dialogue with selected relevant shareholders.

More generally, private meetings between directors and selected shareholders are allowed as long as directors do not discriminate shareholders arbitrarily but resort to specific, objective justifications for the benefit of the company as a whole to support the choice of the investors participating in such meetings.⁴³ This reasoning is coherent with the market sounding regime set out by the MAR, stating under Article 11(2) that disclosure of inside information by a person intending to make a takeover bid for the securities of a company, or a merger with a company, to parties entitled to the securities, shall constitute a market sounding, provided that the willingness of parties entitled to the securities to offer their securities is reasonably required for the decision to make the takeover bid or merger.

But even beyond the paradigm of market soundings,⁴⁴ there are good reasons to believe that ongoing dialogue between the directors and relevant shareholders should be considered beneficial to the company as a whole, and therefore be permitted.⁴⁵ In effect, dialogue-based engagement by institutional investors is *per se* consistent with the interests of the company, since sound investor relations are crucial for publicly listed firms.⁴⁶ Therefore, any deviation from the principle of equal treatment of shareholders for these purposes is not arbitrary, but legitimate, as is also apparent from the SRD II, as well as from the stewardship and corporate governance codes mentioned above, all of which encourage effective boardshareholder dialogue. And, finally, the MAR seems to confirm that a principle of equal information is not absolute in nature also in another

⁴³ See Lombardo and Mucciarelli (2017), p 35.

⁴⁴ The regime for market soundings (Article 11 MAR) indirectly confirms that the MAR leaves sufficient scope for director-shareholder dialogue, insofar as either dialogue does not involve inside information – in which case (arguably, the ordinary case) information can freely be exchanged– or, were inside information to be disclosed to investors participating in dialogue, procedural safeguards modelled on those laid down for market soundings under Article 11 are (voluntarily) put in place in order for the investors participating in the dialogue to benefit from the same protective effect they are granted by Article 11 in the context of market soundings.

⁴⁵ See, for Germany, Hirt et al. (2016), p 737; Fleischer (2009), pp 523-525.

⁴⁶ Hirt et al. (2016), p 737.

respect.⁴⁷ Since both the duty to ensure equal access to information and issuers' disclosure obligation under Article 17 only apply to inside information as defined by Article 7,⁴⁸ the MAR implicitly allows institutional investors to lawfully gain an informative advantage from dialogue with directors, as long as this does not entail the disclosure of inside information.

4 Sustainable Investing as a Driver for Institutional Investor Engagement: the SFDR and the Taxonomy Regulation

One further factor that has proven to support institutional investors' active ownership is sustainable investing, which has gained momentum in more recent years. In particular, climate change has become the most powerful driver of a 'tectonic shift' towards sustainable assets, and of shareholders' demands to investee companies.⁴⁹

As there is an increasing demand for ESG investments,⁵⁰ especially by younger generations,⁵¹ asset managers have been widening their offer of green and sustainable products, most notably including ESG ETFs,⁵² and they are increasingly focusing on sustainability issues in their engagements with portfolio companies.

Sustainable investing has grown strongly in recent years, with asset managers applying, in Europe, 'an ESG investment approach to an estimated total of EUR 11 trillion of assets by the end of Q1 2021'.53 Strikingly,

[b]etween the end of 2010 and the end of 2020, net assets of sustainable equity UCITS increased by 360%, reaching EUR 1.2 trillion by the end of 2020. The share of sustainable equity in total net assets of equity UCITS has also grown, from 22% at the end of 2010 to 29% at the end of 2020.⁵⁴

The rise of sustainable investing has widened the scope of institutions' engagement activities, with stewardship focusing no longer just on the

⁴⁷ Fleischer (2009), pp 512-513.

⁴⁸ See European Court of Justice (2009), para. 48.

⁴⁹ See Fink (2021). See generally Strampelli (2021).

⁵⁰ See, e.g., Global Sustainable Investment Alliance (2021), pp 16-17. See also Standard Chartered Bank (2021), p 3.

⁵¹ See, e.g., Barzuza et al. (2020), p 1285.

⁵² See, e.g., Moisson (2021).

⁵³ European Fund and Asset Management Association (2021), p 19.

⁵⁴ Ibid., p 22.

application of 'governance standards that are considered good practices in the market in which companies are listed' but also on '[e]ncouraging companies to manage environmental and social issues in a sustainable and responsible manner'. Environmental and social concerns are also among the reasons for the growing incidence of activist campaigns in both the United States and Europe, with 'ESG activism ... expanding beyond dedicated funds, ... with several blue-chip activists incorporating ESG into 2021 campaigns'. The surge of ESG investing seems in fact to be capable of broadening the reach of activist intervention and of driving an evolution in the activist landscape. If traditional activism typically leverages business financial underperformance to agitate for strategic changes at targeted companies, a new set of ESG-specialist actors has emerged that emphasize environmental and social factors as a catalyst for change, relatedly to current stock price performance and shareholder return. The surge of ESG in the strategic change, relatedly to current stock price performance and shareholder return.

Indeed, the rapid and ever-increasing rise of sustainable investing 'enables activists to improve perceived ESG weaknesses in businesses but also bolster fundraising by branding themselves as forward-thinking and socially conscious',⁵⁸ and seems likely to further subsidize the interplay between activist and non-activist investors. Especially leading diversified passive asset managers may consider to support activist campaigns and to escalate their stewardship efforts, including by filing proposals to be voted on at shareholders' meetings.⁵⁹ In effect, it is increasingly the case that mainstream institutions back sustainability-related activist proposals with their votes at the shareholders' meeting of portfolio companies. One well-known example is that of Engine No 1, a small hedge fund which, despite its tiny 0.02 percent holding in Exxon shares, gained support from fellow shareholders in its efforts to overhaul the company's board of directors with a view to thus strengthening the company's prospects to abandon fossil fuels.⁶⁰

While institutional investors' sustainability-led active ownership seems to be rapidly turning into a global trend, it is especially strong in Europe,⁶¹ where regulatory action has been stepping up pressure on asset owners and asset managers to integrate climate considerations into both the investment strategies and decisions, and the engagement policies they are required to adopt and publicly disclose, implement and publicly report on under the comply-or-explain rule set by Article 3g of the SRD II. EU legislation is

⁵⁵ Ibid., p 23.

⁵⁶ Lazard (2022), p 23.

⁵⁷ See Lazard (2021), p 21.

⁵⁸ Ibid.

⁵⁹ See Mooney (2021).

⁶⁰ Ibid. See also, for further examples, Ringe (2021a), pp 20-24.

⁶¹ See Morningstar (2021), pp 3-5 and Exhibits 2-3.

clearly directed at enhancing the responsibility (not only of large companies but also) of institutional investors in contributing to address sustainability issues, and has indeed created a normative expectation that institutional investors take an active stance in supporting sustainability with respect to portfolio companies.⁶²

To connect finance with sustainability, since 2018 the Commission has been developing a comprehensive policy agenda on sustainable finance, comprising the Action Plan on Financing Sustainable Growth⁶³ and the development of a Renewed Sustainable Finance Strategy⁶⁴ in the framework of the European Green Deal and the new Strategy for Financing the Transition to a Sustainable Economy.⁶⁵ In its 2018 Action plan the Commission has envisaged to take action aimed at clarifying institutions' duties in relation to sustainability considerations by (i) explicitly requiring institutions 'to integrate sustainability considerations in the investment decision-making process', and (ii) increasing 'transparency towards endinvestors on how they integrate such sustainability factors in their investment decisions, in particular as concerns their exposure to sustainability risks'.⁶⁶

The Commission's intentions rapidly translated into the Sustainable Finance Disclosure Regulation (SFDR), which requires asset managers to explain how they deal with sustainability risks in their investment policies and to describe what effects these risks might have on returns.⁶⁷ As summarized in Recital 12 to the SFDR, financial market participants are required to consider the impacts of investment decisions on sustainability factors in that they

should integrate in their processes, including in their due diligence processes, and should assess on a continuous basis not only all relevant financial risks but also including all relevant sustainability risks that

63 European Commission (2018a).

⁶² See Kelly (2021), pp 18-22.

⁶⁴ European Commission (2018b).

⁶⁵ European Commission (2021a).

⁶⁶ European Commission (2018a), pp 8-9.

⁶⁷ On 6 April 2022 the European Commission adopted Delegated Regulation (EU) 2022/1288 'supplementing Regulation (EU) 2019/2088 of the European Parliament and of the Council with regard to regulatory technical standards specifying the details of the content and presentation of the information in relation to the principle of "do no significant harm", specifying the content, methodologies and presentation of information in relation to sustainability indicators and adverse sustainability impacts, and the content and presentation of the information in relation to the promotion of environmental or social characteristics and sustainable investment objectives in pre-contractual documents, on websites and in periodic reports'.

might have a relevant material negative impact on the financial return of an investment or advice.

Given that 'a sustainability risk means an environmental, social or governance event or condition that, if it occurs, could cause a negative material impact on the value of the investment',⁶⁸ in the eyes of the EU legislator, enhanced pre-contractual and ongoing disclosures about the integration of sustainability risks into the investment process, alongside the adoption of transparent written policies for integrating sustainability risks, could direct the assets of end-investors towards institutional investors most conscious of and active on sustainability risks, thereby ultimately channeling investments towards environment-friendly or socially committed businesses.⁶⁹

In line with this policy goal, and based on the consideration that, indeed, '[s]ustainable products with various degrees of ambition have been developed to date',⁷⁰ sustainability-related pre-contractual disclosures required by the SFDR vary depending on whether or not the investment products marketed by a given institution do have stated sustainability or ESG ambitions.

Thus, Article 6 SFDR sets the minimum standard for pre-contractual transparency required of any investor by imposing on all market participants an obligation to disclose the integration of sustainability risks, whether material or not, and their likely impacts on the returns of the financial products they make available. Article 8 then requires institutions that sell products which *promote* environmental and/or social characteristics, among others, to additionally specify how they will promote such characteristics - whether or not an index has been designated as a reference benchmark - and how the companies they are invested in follow good governance practices. In turn, Article 9 requires market participants that sell products which have as an objective a positive environmental or societal impact, and for which an index has been designated as a reference benchmark, to provide pre-contractual information on how the designated index is aligned with that objective, and an explanation as to why and how the designated index aligned with that objective differs from a broad market index. Where no index has been designated as a reference benchmark, the information to be disclosed needs to include an explanation on how that objective is to be met. Further still, information due under Article 9 includes an indication of where the methodology used for the calculation of the indices and the benchmark adopted is to be found.

⁶⁸ Article 2 (22) SFDR, and Recital 14 to the SFDR.

⁶⁹ See Recital 8 to the SFDR.

⁷⁰ Recital 21 to the SFDR.

In addition to pre-contractual disclosures, further information is required by Article 11 SFDR of Article 8 and Article 9 investors to be disclosed in ongoing annual reports, explaining how the sustainability characteristics, or the sustainable objective, of the financial products are met.

As a consequence of such varied disclosure requirements, the SFDR has 'de facto split the EU fund universe into three categories':⁷¹ conventional Article 6 funds with no stated sustainability or ESG ambition, required to disclose how they integrate sustainability risks into the investment process; Article 8 funds that, among other characteristics, promote sustainability characteristics [also called 'light green' funds], and Article 9 funds that do have a sustainability objective ['dark green' funds]. While Article 9 funds are sometimes referred to as 'impact funds', the term 'sustainable funds' more closely aligns with the wording of Article 2(17) SFDR.⁷²

It should be noted that the three-pronged classification of the investment products set by the SFDR was not intended to create product labels,⁷³ but rather was motivated by the willingness to bring some order in the high-rocketing but quite disorderly universe of investment products featuring self-branded – and not always truthful – ESG characteristics and avoid, or at least restrain, greenwashing. Interestingly, following on from the sustainability disclosures rules implemented in March 2021, after noting that Europe accounted for 88 per cent of the global market in ESG products and finding 'ambiguous language' in legal filings, Morningstar removed the sustainable investment label from more than 1,200 funds (totalizing \$1.4tn in assets) previously included in its list, 'including many that listed ESG criteria to self-classify as promoting environmental and/or social characteristics under European disclosure rules', mostly under Article 8 SFDR.⁷⁴ In and for itself alone, such adjustment provides a clear picture of the scale and seriousness of the challenge facing both end-investors and

⁷¹ European Fund and Asset Management Association (2021), p 19.

⁷² According to which a 'sustainable investment' is 'an investment in an economic activity that contributes to an environmental objective, as measured, for example, by key resource efficiency indicators on the use of energy, renewable energy, raw materials, water and land, on the production of waste, and greenhouse gas emissions, or on its impact on biodiversity and the circular economy, or an investment in an economic activity that contributes to a social objective, in particular an investment that contributes to tackling inequality or that fosters social cohesion, social integration and labour relations, or an investment in human capital or economically or socially disadvantaged communities, provided that such investments do not significantly harm any of those objectives and that the investee companies follow good governance practices, in particular with respect to sound management structures, employee relations, remuneration of staff and tax compliance'.

⁷³ European Fund and Asset Management Association (2021), p 19.

⁷⁴ Quinio (2022).

regulators vis-à-vis institutional investors and asset managers in finding common ground on what constitutes green or sustainable investments.

As mentioned earlier, one further pillar of the Commission's Action Plan on Financing Sustainable Growth is the Taxonomy Regulation, by which the EU provides a taxonomy of the activities that can be considered sustainable as regards, in a first step,⁷⁵ the issue of climate change, with institutional investors that sell financial products marketed environmentally sustainable required to disclose how and to what extent these financial products are actually invested in environmentally sustainable activities as per the taxonomy. By providing a unified classification system for economic activities that can be considered (environmentally) sustainable based on performance thresholds for six key environmental objectives (see Article 9 of the Taxonomy Regulation) and requiring that sustainable activities substantially contribute to at least one of these without harming any other (see Article 10 of the Taxonomy Regulation), the EU taxonomy is explicitly (and more strongly than the SFRD) intended to prevent market participants from engaging in greenwashing when disclosing their sustainability footprint, and to 'enhance investor confidence and awareness of the environmental impact of those financial products or corporate bonds'.76

While only applying to SFDR Article 8 and Article 9 investment products, and large companies, the EU taxonomy is however likely to indirectly impact other financial market participants and other issuers seeking to attract sustainability-focused capital as well, as it may be expected to drive broader market expectations of disclosure. Encouraging actors other than those subject to the Taxonomy Regulation to voluntarily comply with this Regulation is, in fact, an outcome explicitly considered by the EU legislators, who point out that voluntary disclosures aligned with the obligations set by the Regulation

will not only help financial market participants and other relevant actors on the financial markets to easily identify which economic operators carry out environmentally sustainable economic activities, but will also make it easier for those economic operators to raise funding for their environmentally sustainable activities.⁷⁷

16

⁷⁵ See Recital 6 to the Taxonomy Regulation. Disclosure requirements associated with climate change mitigation and adaptation objectives apply from 1 January 2022; those set for all other environment-related objectives apply from 1 January 2023. In turn, social and governance objectives are expected to be proposed as additions to the existing framework in the coming year.

⁷⁶ Recital 11 to the Taxonomy Regulation.

⁷⁷ Recital 15 to the Taxonomy Regulation.

5 The Issuer Side of Sustainability: from the NFRD to the CSRD, Through to the CSDD

One fundamental tool for meeting the disclosure obligations imposed on institutional investors by the SFDR, in alignment with the EU Taxonomy, are – alongside engaging with the board and major external stakeholders – issuers' sustainability reports, disciplined by Directive 2014/95/EU (Non-Financial Disclosure Directive – NFRD) which requires certain large companies to provide non-financial reports alongside the yearly financial reports, as a form of disclosure of information on the way they manage social and environmental challenges. The relevance of issuers' non-financial reports for institutional investors' disclosures is explicitly acknowledged by the SFDR. As stated by Article 11(3) SFDR with respect to Article 8 and Article 9 funds, for the purposes of their ongoing reports, financial market participants may resort to corporate information included (in management reports required by the Accounting Directive and) in non-financial statements. As is quite evident,

[w]ithout sufficient, reliable and comparable sustainability-related information from investee companies, the financial sector cannot efficiently direct capital to investments that drive solutions to the sustainability crises we face, and cannot effectively identify and manage the risks to investments that will arise from those crises.⁷⁸

Moreover, the Taxonomy Regulation requires companies under the scope of the NFRD to disclose their alignment with activities that qualify as being environmentally sustainable according to the EU taxonomy. Hence, availability of reliable, comparable and relevant information on sustainability risks, opportunities and impacts is essential for the successful implementation of the objectives of both the SFRD and the Taxonomy regulation.⁷⁹

The developments in the EU regulatory framework since the adoption of the NFRD, and dissatisfaction with the actual quality of non-financial information provided under the NFRD despite non-binding guidance provided by the Commission in 2017 and 2019,80 are among the reasons why, based on the review clause in the NFRD, the EU is heading towards amending the existing non-financial reporting requirements by adopting a

 $^{^{78}}$ European Commission, Guidelines on non-financial reporting: Supplement on reporting climate-related information, 20 June 2019, 2019/C 209/01, OJ C 209/1, C 209/2.

⁷⁹ European Commission (2021b), p. 5-6.

⁸⁰ European Commission (2021c); European Commission (2021d).

new Corporate Sustainability Reporting Directive (CSRD). The CSRD not only extends the scope of application of the NFRD to all large companies and all publicly listed firms (except for listed micro enterprises) and requires audit assurance of reported information, but, most notably, introduces a requirement for issuers to report according to mandatory EU sustainability reporting standards to be developed around the concept of double materiality, hence considering materiality of ESG issues from both a financial and an impact (non-financial) perspective.81 Under the CSRD, addressee firms are thus going to be required 'to report about how sustainability issues affect their business and about their own impact on people and the environment', with both dimensions (financial, according to an outside-in perspective, and impact, according to an inside-out perspective) extending beyond matters that are within the direct control of the reporting entity, since they include its down- and upstream value chain. According to the Commission, the adoption, by means of delegated regulations, of mandatory EU sustainability reporting standards - the first set of which is to be delivered by 30 June 2023 - derives from the need to ensure comparability and completeness of the information disclosed, enable the audit and digitalization of sustainability reporting, and facilitate its supervision and enforcement.82 Moreover, '[n]o existing standard or framework satisfies the Union's needs for detailed sustainability reporting by itself', also considering the need for such standard to align, amongst others, with the disclosure requirements laid down in the Taxonomy Regulation and the underlying indicators and methodologies set out in the delegated acts adopted under the SFDR.83 From the standpoint of the main users of sustainability reporting, i.e., of institutional investors, the proposed adoption of mandatory, double-materiality principled reporting standards is of paramount importance, since it is obviously bound to heavily impact their processes and decisions.

The broad framework drawn by the EU in regard to sustainability includes one further piece of legislation which targets corporate issuers concerning sustainable corporate governance. Following on from its 'Sustainable Corporate Governance' Initiative aiming 'to improve the EU regulatory framework on company law and corporate governance', on the one hand, by 'enabl[ing] companies to focus on long-term sustainable value creation rather than short-term benefits', and, on the other hand, 'to better align the interests of companies, their shareholders, managers, stakeholders

_

⁸¹ European Commission (2021b). The text of the Directive following approval by the European Parliament on 10 November 2022 is available at https://www.europarl.europa.eu/doceo/document/TA-9-2022-0380_EN.html#title2 (accessed 22 November 2022).

⁸² Ibid., Recital 32.

⁸³ Ibid., Recitals 33 and 35.

and society' and 'help companies to better manage sustainability-related matters in their own operations and value chains as regards social and human rights, climate change, environment, etc.',84 in early 2022 the European Commission adopted a Proposal for a Directive on Corporate Sustainability Due Diligence (CSDD).85

The CSDD Proposal eventually did not follow up on the issue of curbing short-termism, at least not in regard to investor-driven, tradingmotivated short-termism allegedly resulting in undue pressure on company directors to deliver short-term financial results and resort to management strategies aimed at sustaining share prices in the short run. To be true, no single provision of the CSDD Proposal actually refers to addressee companies' shareholders, while granting their stakeholders which, according to Article 3(n) of the Proposal, do not (explicitly) include the shareholders⁸⁶ - a series of prerogatives ranging from that of being consulted when identifying the company's actual and potential human rights and environmental adverse impacts and when developing a prevention action plan as well as a corrective action plan,87 to the right to submit complaints in case of legitimate concerns regarding those potential or actual adverse impacts, to submit substantiated concerns to any supervisory authority when believing that a company is failing to comply with the Directive, and to provide inputs relevant to setting up the company's due diligence policies,88 through to the right to claim compensation for damages suffered as a consequence of the company's failure to comply with its due diligence obligations.89

Still, the provisions of the CSDD affect institutional shareholders indirectly. Specifically, the proposed rules that require large companies and those operating in certain high-impact sectors, including non-EU companies, to identify and, where necessary, prevent, end or mitigate adverse impacts of their activities on human rights and on the environment, will provide investors with some assurance, additional to that stemming from corporate sustainability reports, that actual or potential portfolio companies do engage in managing ESG risks, thus strengthening the overall reliability of the sustainability information investors use in the investment process and the associated disclosures, as mandated under the SFDR. As

⁸⁴ European Commission (2020).

⁸⁵ European Commission (2022a).

⁸⁶ Based on Article 3(n) of the CSDD Proposal, "stakeholders" means the company's employees, the employees of its subsidiaries, and other individuals, groups, communities or entities whose rights or interests are or could be affected by the products, services and operations of that company, its subsidiaries and its business relationships'.

⁸⁷ See Articles 6(4), 7(2)(a) and 8(3)(b) of the CSDD Proposal.

⁸⁸ See Articles 9, 19 and 26 of the CSDD Proposal.

⁸⁹ See Article 22 of the CSDD Proposal.

pointed out by the Commission, the proposed CSDD will complement the framework set out by the NFRD and the CSRD

by adding a substantive corporate duty for some companies to perform due diligence to identify, prevent, mitigate and account for external harm resulting from adverse human rights and environmental impacts in the company's own operations, its subsidiaries and in the value chain.⁹⁰

6 Uncertainties Surrounding Institutional Investors' Walk to Sustainable Finance

In spite of the growing practical relevance of active share ownership generally, and the increasing reliance made by the EU on institutions' stewardship to support the transition toward sustainability and, particularly, to counter climate change, whether institutional investors and asset managers are motivated to effectively play such crucial a role remains controversial. Even assuming they were motivated, and were committed to not just cosmetically contributing to mitigating the ever-increasing, and no more just looming but already materializing, systemic risks posed by global environmental and social issues, still there are limitations to the reasonable reach of investor action in face of the scale of the challenges at stake. Limitations not only derive from the deficient incentives structure and the collective action issues that are typical of asset managers. They also depend on external factors not in the asset manager's control, such as voluble endinvestor preferences and priorities, and availability of better and more reliable ESG data and information to drive institutions' assessments, and the cost thereof. Skepticism concerning the capacity of institutional investors and asset managers to be active owners in regard to sustainability issues seems to be justified on account of evidence referring to institutions' actual voting behavior at portfolio companies.

6.1 Resource and Cost Constraints to Active Stewardship, and the Problem of Collective Action

The actual capacity of institutional investors to provide tangible incentives for action by investee companies that are informed by sustainability objectives and to restrict the pursuit of different short-term objectives is dependent, first and foremost, on the scale of the resources dedicated to achieving those objectives. The evidence shows that the stewardship teams

⁹⁰ European Commission (2022a), p 4.

of major institutional investors are significantly smaller than the size of the task they are entrusted with. For example, in BlackRock's case it is estimated that investment in stewardship activity totals around 13.5 million dollars, which is equivalent to around 0.15% of the (estimated) asset management fees collected. In addition, stewardship teams appear to be largely understaffed relative to the number of companies covered by stewardship initiatives. Given their limited size, stewardship teams are unable to carry out specific checks in regard to a large number of portfolio companies, regarding which they do not actively engage but apply predetermined voting policies, or rely on the assistance of proxy advisors, the accuracy of whose analyses is, as is known, a matter of debate.

More so, engaging on an ever-expanding array of topics indeed requires asset managers to dedicate more and more time and resources to stewardship, thus exacerbating the cost- and incentive-issue. As the scale of the challenges associated is 'likely to favor larger players', 94 and in any case imposes to set priorities based on which to select which companies to engage with, and on which topics, medium and small-sized portfolio companies, and those in which stakes are less relevant, are likely to escape investor scrutiny to a larger extent, irrespective of the relevance of any ESG concern they may pose. The very same scale considerations are also likely to further contribute to the factual concentration of corporate power in the hands of a few larger universal owners. If, due to cost considerations, size is among the determinants of stewardship, the broadening topics to be covered erect for small institutional investors and asset managers higher barriers to engagement than those imposed on large investors. 95 Both these factors are to be reckoned with when assessing the European ambitions related to sustainable finance.

Beyond the issue of size, resource and cost constraints are particularly pronounced where passive investment prevails. In fact, costs associated with stewardship impinge much more significantly on asset managers' income, as passive funds have much lower fees. Performance fees do not usually apply for passively managed vehicles, but rather only subscription fees proportional to the amounts invested in the fund; therefore, the incentive for managers to allocate funds to stewardship activities with the aim of improving the fund's return is limited.⁹⁶ Indeed, passively managed investments

⁹³ For example, BlackRocks' stewardship teams currently comprises 46 persons: BlackRock (2020), p 5. See, e.g., Franks (2020), p 266; Sharfman (2022b), p 15.

⁹¹ See, e.g., Bebchuk and Hirst (2019), pp 2077-2080.

⁹² Ibid.

⁹⁴ Goldman Sachs Equity Research (2019), p 5.

⁹⁵ See Ringe (2021b), p 95.

⁹⁶ See Kahan and Rock (2020), pp 1795-1797.

exist in a super competitive industry with extremely low management fees, providing [even the largest global players dominating passive investing] with very little ability to spend resources on becoming informed about portfolio companies. In sum, the Big Three are not being paid to be informed, only to do as much as they can to serve the interests of beneficial investors when they vote and engage on their behalf.⁹⁷

One further factor that adversely affects investment in stewardship is the fact that highly diversified investors face significant collective action problems. This is especially the case for passively managed funds because, as each fund tracking the same index holds the same stocks in the same proportion, funds managed by other index fund managers will capture exactly the same returns from the stewardship activity. To be true, the scale of such disincentive is reduced, at least to some extent, in the case of ESG funds, and even more so of customized ESG funds, since sectoral passive funds and personalized index funds, unrepresentative of the broader market, are created around particular, smaller or fund-specific benchmark indices actually based on active-like investment strategies which lead to more concentrated portfolios. 99

However, irrespective of the scale of collective action problems, there still remains that index fund managers are able to capture only a small fraction of the benefits from stewardship, given the very low fees that they charge. Therefore, sensitivity to the free-rider problem is high, particularly in contexts where competition for investor capital is high, as is the case within the asset management industry, where fund managers compete to attract assets under management based on performance relative to alternative investment opportunities. At the same time, according to some scholars, leading institutional investors have strong incentives to be deferential to the preferences of portfolio companies' managers as they have significant business ties with the companies in which they hold positions. 102

However, these lines of argumentation are controversial, since other scholars contend that leading passive, highly diversified investors do have

98 See Bebchuk and Hirst (2022), p 41.

⁹⁷ Sharfman (2022).

⁹⁹ See Robertson (2019), p 795 et seq; Mahoney and Robertson (2021), p 311 et seq.

¹⁰⁰ Bebchuk and Hirst (2022), p 42.

¹⁰¹ Coates, IV, and Glenn Hubbard (2007), p 5.

¹⁰² Bebchuk and Hirst (2022), p 47.

incentives to invest in stewardship and, specifically, to actively engage with portfolio companies.¹⁰³

First, in order to reduce the comparative advantage that competing actively managed funds enjoy vis-à-vis passive funds on account of their ability to trade, passive funds need to exert their voice to improve the corporate governance and performance of investee companies and prevent asset outflow. 104 Such reasoning seems to be reinforced by the tendency of institutional investors to incorporate ESG factors into their investment and stewardship guidelines. As clearly stated by Larry Fink, leading asset managers focus on sustainability not because they are environmentalists, but because they must act in the interest of their end clients, whereas tackling portfolio-wide externalities or sustainability-related systematic risk can improve risk-adjusted portfolio returns. 105 Along the same lines, it is also theoretically plausible that the EU taxonomy, and mandatory investor disclosures drafted against the taxonomy, may spur competition for sustainability-minded end-investors based on the degree of taxonomy alignment of the funds managed by them, thus prompting institutional investors to enhance voice as the primary strategy by which to achieve higher levels of taxonomy alignment and more credibly attract endclients.106

However, contrary to this view, it should be noticed that 'removing systematic risk from the portfolio, whether caused by climate change or social risk, is likely to be a costly endeavor' that does not lend itself to a low-cost stewardship approach, and, as the benefits of any market-wide intervention undertaken would be shared by competing institutions, there is 'a strong incentive not to undertake them and instead take a free ride'. ¹⁰⁷ In effect, for portfolio primacy to be effective in regard to voting and engagement as a way to maximize end-investor returns across the portfolio, asset managers need to be informed about portfolio firms: which, precisely, is contested. Hence, it is argued, being uninformed, 'the only practical way for the Big Three to implement portfolio primacy is to implement a general policy of deferring to board authority at their portfolio companies' in spite of significant bias that may exist in some board recommendations. ¹⁰⁸

Second, given their interest in preventing asset outflow and attracting new clients, there is growing reputational and regulatory pressure for leading fund managers – chiefly the largest passive fund managers – to be

¹⁰³ See Condon (2020), p 1.

¹⁰⁴ See Fisch et al. (2019); Kahan and Rock (2020), pp 1795-1797.

¹⁰⁵ Fink (2022). See also, amongst others, Gordon (2022), pp 2-3; Azar et al. (2021), p 675.

¹⁰⁶ See Pacces (2021), p 12 et seq.

¹⁰⁷ Lund (2022), pp 12-13 and 26.

¹⁰⁸ Sharfman (2022b), pp 24-25.

active monitors.¹⁰⁹ It can hence be argued that, faced with mounting pressure from a variety of stakeholders, asset managers have no option but to increase their level of engagement with portfolio firms to demonstrate their commitment in encouraging positive change.¹¹⁰ However, it is also credible that creating the appearance of active ownership will help fund managers to win over clients. Therefore, fund managers may see corporate engagement 'as a branding or marketing tool that provides them with another dimension on which to compete for assets'.¹¹¹

6.2 Dependency on End-Investor Preferences and Priorities

Also due to the fact that ESG investments do not have a very long record, evidence is still mixed concerning whether or not sustainable investing holds promise on delivering financial returns that are competitive vis-à-vis non-ESG investing. Asset managers' readiness to direct investment to sustainable businesses remains open to discussion, as badly performing ESG funds might very well induce beneficial investors to switch their preferences towards less sustainable but more profitable investments. Simply put, markets' cyclicality does not necessarily, nor easily, match with investor preferences and hence with ESG challenges, that are structural by nature, since it is still unclear whether investors are actually willing to forego profits when investing in ESG products.

After all, as acknowledged by BlackRock, the success of investment strategies incorporating sustainability considerations is due, first of all, to the belief that, by mitigating systematic risk at portfolio level, these strategies are likely to achieve higher returns at reduced risk. However, studies do not unequivocally support this view. A number of empirical analyses have not identified any increase either in the returns at firms that enact sustainability-informed strategies or in the returns at portfolios that include companies with better ESG ratings and performance. And,

¹¹¹ Fisch et al. (2019), p 13; Kahan and Rock (2020), pp 1799-1800.

¹⁰⁹ See, e.g., Kahan and Rock (2020), p 1797.

¹¹⁰ See Biov (2017).

¹¹² See Boffo and Patalano (2020), p 3; Bhagat (2022).

¹¹³ See Somerset Webb (2022).

¹¹⁴ See Zetsche and Anker-Sørensen (2022), p 65.

¹¹⁵ BlackRock Investment Institute (2020), p 3.

¹¹⁶ Davies (2020), p 22; Zetsche and Anker-Sørensen (2022), p 64.

¹¹⁷ See Lopez de Silanes et al. (2019), p 2; Bolton and Kacperczyk (2021), p 517; Demers et al. (2021); Hartzmark and Sussmann (2019); Raghunandan and Rajgopa (2022), p l. However, as a demonstration of how arguable the conclusions are in this area, it is interesting to note that indices based on ESG parameters appear to have better withstood the financial market crisis triggered by the spread of the Coronavirus in February and March 2020. See Tett et al. (2020).

importantly, it should be noted that, based on a global survey, active asset managers 'expressed commitment to ESG goals in their investing and as a priority for their portfolio companies', but, at the same time, 'most (81% of) respondents expressed reluctance to take a hit on their returns exceeding 1 percentage point in the pursuit of ESG goal'.118 Further, composite ESG scores seem to be a strong driver for institutional investors when picking stocks to be added to portfolios, with the governance component having the highest impact on holdings. The prevalence of the G dimension of ESG scores in driving firm selection shows that 'it is through investing in forms with high governance quality, that investors are able to effectively drive increased E and S performance',119 namely by engaging with investee companies. However, ESG scores are not a driver for the size of ownership stakes, suggesting that large investors are much less sensitive to ESG considerations. Where looking at the individual components of ESG scores, the governance component has been found to more robustly correlate with financial performance, in terms of reduced volatility and systemic risk, than the E and S components. 120

Moreover, if more robust evidence were to be found that ESG funds do not deliver better ESG performance either, then one should conclude that 'funds investing in companies that publicly embrace ESG sacrifice financial returns without gaining much, if anything, in terms of actually furthering ESG interests'. The latter conclusion is however essentially tentative, since, as cautioned by EU institutions, availability of information that may be used to assess non-financial performance of sustainable investment products is 'largely inadequate'. As 'fund performance associated with ESG varies considerably, depending on the scores used and the analytical approaches employed', amore research is needed to assess the extent to which ESG scoring methodologies can affect the measure of performance. At any rate, both these issues need to be accounted for, since they might possibly adversely affect end-investor inflow to, and asset managers' interest in, ESG pooled vehicles.

6.3 External Limitations to Institutional Investors' ESG Investing and Stewardship

¹¹⁸ Chalmers et al. (2021), pp 2 and 4.

¹¹⁹ Lopez de Silanes et al. (2022), p 17.

¹²⁰ Ibid., pp 23-24.

¹²¹ Bhagat (2022).

¹²² European Securities and Markets Association (2022), p 21.

¹²³ Boffo and Patalano (2020), p 35.

¹²⁴ Ibid, p 37.

One serious challenge investors are faced with when evaluating ESG priorities is the lack of quality and consistency of the information made available to them by both issuers' non-financial reports and third-party data providers' ESG benchmarks and ratings. Also due to the absence of consistent and comparable issuer non-financial disclosures, ESG ratings and data products have grown into essential tools within the ESG ecosystem, and, following increased demand, the role and influence of private providers of ESG ratings and data products have grown significantly. ESG ratings and data products are not flawless, however, and do not typically fall within the remit of regulation. As a result, deficiencies in the quality of the relevant information makes it hard for investors to differentiate between companies on ESG-related performance, and try to shift capital towards firms credibly making progress on the way to, e.g., net zero.

While the CSRD is intended to tackle the issue of insufficient relevance, reliability, and comparability of corporate sustainability reporting, still impaired ESG investors' assessments caused by the adoption of differing sustainability reporting frameworks will not be overcome unless meaningful international convergence of such standards is achieved.

But, on top of issuer disclosures, differing, and opaque, ESG indexes and ratings sold by the index and ESG rating providers industry add one further layer of complexity to investors' assessments. As is well known, asset managers are heavy benchmark users, both in the case of index funds and ETFs – where benchmarks are used as a target for index tracking funds – and in the case of the evaluation of an active manager's performance – where the fund performance is measured against an index or a set of indices. Hence, the use made by asset managers of ESG information is not only direct – through access to issuer-disseminated non-financial disclosures – but also indirect, through ESG ratings commercially available or, to a lesser extent, that are built by the asset manager itself.

Rating practices, in terms of determining which data to include, and how to weigh metrics in terms of materiality, and layering subjective judgment as to absolute and relative scores within and across industries, are very wide-ranged, and third-party index providers and rating agencies develop and sell ESG ratings and indices based on intrinsically different data and methodologies.¹²⁸ Ratings, data and indices used by market participants to identify and assess companies that adopt better ESG practices – either to make active investment decisions through a selected

¹²⁵ See, e.g., PricewaterhouseCoopers (2021), pp 4-5.

¹²⁶ See International Organization of Securities Commissions (2021), p 5.

¹²⁷ Ibid., p 1.

¹²⁸ See Boffo and Patalano (2020), p 21.

approach relative to an index, to structure investment portfolios aligned with a given index, or to take stewardship action – are not only largely varied and not comparable, but also opaque, due to a fundamental lack of transparency in the underlying methodologies.¹²⁹

Hence, the problem of non-consistent frameworks for corporate non-financial disclosures, and the underlying diverging concepts of materiality, couples with that of divergent, and opaque, ESG ratings and indices. ¹³⁰ In particular, low correlation among the scores assigned by different providers to the same companies, and the lack of transparency in how such scores are developed, make it hard for investors to resort to yet essential, and reliable, information they need to properly perform sustainability assessment. Tellingly enough, the lack of transparency around external ESG rating methodologies is among the reasons why users are encouraged to build proprietary rating methodologies. ¹³¹ Commendable as it may be, this option is however factually limited to the largest asset managers, since 'producing proprietary internal ESG ratings may not be feasible or cost effective for small or medium sized asset managers'. ¹³²

In addition, with ESG scores largely dependent on the methodologies used by providers – either ESG raters or portfolio managers themselves – 'the extent to which end-investors can be assured that ESG investing provides enhanced returns or aligns with any particular societal values is dubious', ¹³³ which, in turn, is likely to 'confuse end investors and undermine their trust and confidence in sustainable financial products'. ¹³⁴ It plainly follows that

[g]iven the influence of ratings providers, the differences in ratings methodologies, and their level of transparency in final rating decisions that also incorporate qualitative judgments, are critical to understanding the resilience of the ESG financial intermediation chain.¹³⁵

Further concern associated with ESG ratings and indices is, alongside high levels of market concentration around a few global players, the potential for conflict of interests deriving from providers offering companies other types of services (as ESG ratings providers can assume different roles, such as consultant, data provider, or rating provider,

¹³⁰ See Chiu (2022), p. 105 et seq.; Zetsche and Anker-Sørensen (2022), pp 70-71.

¹²⁹ Ibid.

¹³¹ International Organization of Securities Commissions (2021), p 27.

¹³² Ibid.

¹³³ Ibid, p 29.

¹³⁴ European Fund and Asset Management Association (2022a), p 3.

¹³⁵ Boffo and Patalano (2020), p 22.

representing the interests of both companies and investors), notably with regard to their ESG performance. Additional concern derives from uneven covering of the products offered (with certain industries, companies or geographical areas benefitting from more coverage than other firms), ¹³⁶ and the rising, and excessive, cost of ESG data services which, associated with the lack of pricing transparency, is 'particularly detrimental to smaller firms with fewer resources and bargaining power and ultimately affects end investors as potentially these costs may be passed on to them', and can possibly be in breach of 'the non-discrimination and cost-based/cost-related principles laid down in the various national regulations'. ¹³⁷

While in the EU the regulatory framework for benchmarks may be relevant for certain providers of benchmarks with an ESG or climate dimension, it is not directly relevant for the broad scope of ESG ratings and data products.¹³⁸ ESG indices fall within the scope of the Benchmarks Regulation,139 but ESG ratings and assessments do not. Yet ESG benchmarks are not independent of ESG ratings. As highlighted by the ESMA, the very low levels of correlation that different ESG ratings display make the construction of ESG benchmarks problematic 'with the choice of ESG rating provider significantly impacting the constituents of those indices'.140 Moreover, as 'companies in highly polluting industries can obtain high environmental scores from some ESG rating providers', 141 there is the risk, ultimately, that capital allocation aligned with sustainability objectives is impaired, undermining the sustainability-related endeavors of both sustainability-minded end-investors and the EU. Therefore, in light of the ever-growing size of the industry, and interest in ESG benchmarks, ensuring the credibility of ratings is necessary.

Having identified a number of key points for consideration, early in 2021, the ESMA urged the European Commission to take appropriate regulatory steps to ensure the quality and comparability of ESG ratings and assessment tools, following which the Commission launched a consultation, the results of which may induce the Commission to actually

¹³⁶ See International Organization of Securities Commissions (2021), p 10 et seq.

¹³⁷ See European Fund and Asset Management Association (2022a), pp 3 and 6.

¹³⁸ International Organization of Securities Commissions (2021), p 14.

¹³⁹ Regulation (EU) 2016/1011 of the European Parliament and of the Council of 8 June 2016 on indices used as benchmarks in financial instruments and financial contracts or to measure the performance of investment funds and amending Directives 2008/48/EC and 2014/17/EU and Regulation EU (No 596/2014).

¹⁴⁰ European Securities and Markets Authority (2021), p 2.

¹⁴¹ Ibid.

¹⁴² Ibid.

take regulatory action in the field,¹⁴³ also following up on the recommendations issued by the IOSCO.¹⁴⁴

6.4 ESG Voting in Practice

Some skepticism concerning the capacity and willingness of institutional investors to be active owners in regard to sustainability issues seems to be justified on account of evidence referring to institutions' actual voting behavior at portfolio companies. Statistics concerning ESG engagement and, specifically, voting on environmental and social matters, depict a picture that would appear not to be perfectly aligned with institutions' claimed ambitions. Based on an analysis of how 65 of the world's largest asset managers voted in 2021 across 146 social and environmental resolutions globally, the actual reach of global asset managers' active ESG share ownership still remains open to discussion, especially as far as the very largest players are concerned.¹⁴⁵

First, voting support to E&S resolutions appears to be dominated by European asset managers, whose 'for' votes outpace on average those of US-based institutions by 25 per cent in frequency (64 and 39 per cent respectively). Arguably, the favorable regulatory environment in the EU acts as a driver for European investors' enhanced positive voting behavior compared to US peers.

Second, the six largest asset managers in the world – BlackRock, Vanguard, Fidelity Investments, State Street Global Advisors, Capital Group, and J.P. Morgan Asset Management – each supported fewer than 40 per cent of E&S resolutions they voted on. And, interestingly enough, they all showed a tendency to vote more conservatively than recommended by major proxy advisory firms, considering that ISS recommended backing 75 per cent of the proposals assessed, and Glass Lewis 44 per cent. While not being a feature exclusive of the very largest asset managers (68 per cent of the institutions considered actually shared this trend of supporting fewer proposals than recommended by ISS, or 23 per cent where considering Glass Lewis' recommendations 148), such tendency is however particularly evident for US-based institutions, whose voting records seem to be at odds with their claims of being at the forefront of ESG engagement. As has been noticed, an additional 10 per cent of the resolutions examined for 2021

¹⁴⁷ Ibid., p 21.

¹⁴³ European Commission (2022b).

¹⁴⁴ International Organization of Securities Commissions (2021), Chapter 5.

¹⁴⁵ See ShareAction (2021).

¹⁴⁶ Ibid., p 13.

¹⁴⁸ Ibid., p 22.

¹⁴⁹ See Lafarre (2022), p 6.

would have achieved majority support had one or more of the Big Three voted in favour instead of against.¹⁵⁰ Where the size of the assets under management and the potential of influence these players can wield over corporate behavior are accounted for,¹⁵¹ these findings shed some shadows on the reasonableness of the EU institutions' heavy reliance on institutional investors' contribution to the implementation of the sustainable finance agenda.

That holds true, at least trend-wise, even if proper consideration is given to the fact that the proportion of passive investments relative to the overall size of assets under management in the EU is lower, and perhaps even significantly lower, than in the United States, and that, correspondingly, common ownership in continental Europe is less significant.¹⁵² The power, and problems, of highly diversified institutions, in particular index fund managers such as the prototypical Big Three, may at present be less enhanced in Europe than in the United States -an issue to be factored in, and further explored, by EU regulators when shaping any policy and intervention. Yet, they cannot be underestimated. First, at least at some EU firms, particularly large dispersed firms which are typically represented in many indices, diversified ownership and the stakes collectively held by index funds are notable. 153 Second, alongside large universal owners such as the Big Three, other asset managers are passive investors, or quasi-indexers. Finally, where looking at the outgrowth of index funds and ETF-investing in the EU and their speedy pace, and considering that 'the main reason behind this growth is the lower cost of passive funds and ETFs',154 the common expectation is for such evolutionary trend to further strengthen.

It may very well be, of course, that failure to support E&S proposals is motivated by the willingness not to oppose the management where the investor believes that sufficient progress is being made by the company in regard to the relevant issue. This is particularly true where such belief is based on previous behind-the-scenes engagements with the issuer. After all,

¹⁵² See Rosati et al. (2020), pp 52-53. Less pronounced (but still meaningful, at least among the largest issuers with no controlling shareholders: see Steuer (2022), p 29) levels of common ownership in the EU relative to the United States seem consistent with both the facts that the passive investment segment in the EU is overall less pronounced than in the United States (see European Securities and Markets Authority (2022), p 13; European Fund and Asset Management Association (2022b), p 42), and that large non-institutional blockholders, including controlling stockholders, are common at European listed firms, whereas they are not in the Unites States (see De La Cruz et al (2019), p 11).

 $^{^{150}}$ ShareAction (2021), p 25. These data are consistent with Griffin (2020), pp 411, 422, 444. See also Posner (2021).

¹⁵¹ ShareAction (2021), pp 23-24.

¹⁵³ See Steuer (2022), p 27.

¹⁵⁴ See European Fund and Asset Management Association (2021), p 46.

voting against the management is *per se* a confrontational stewardship tool that stewardship principles recommend to resort to by way of escalation. Still, the imbalance between voting behaviour and divergent proxy advisors' recommendations, coupled with time- resource- and cost-related constraints on institutions' stewardship, as well as conflicted business models that are typical for large asset managers, remains remain striking.

The fact that most asset managers' E&S voting behaviour tends to remain the same over the years, and that the overall proportion of 'for' votes only shows slight increases, 155 coupled with further findings that some (European) players fail to exercise more than 40 per cent of their voting rights – in spite of being often a member of the Climate Action 100+ initiative and a signatory to the Principles of Responsible Investment (PRI) 156 – and that the vast majority of asset managers are reluctant to file, or co-file, E&S shareholder proposals at portfolio companies, 157 may be interpreted as a signal that investors are, at a minimum, too slow at adapting to the urgency of the challenges posed by E&S risks on the way they manage assets.

Even more so, investors' adaptation to sustainable investing and to the needs of booming sustainability-aware end-clients is slow where the socalled impact perspective of investment management is accounted for. As further inquiries suggest, while asset managers are increasingly considering the implications of ESG issues for financial performance where making investment decisions, a conceptual shift towards considering the real-world impact of their investments on the environment and society, and more comprehensively integrating systemic ESG risks into the investment process, seems far from being achieved in spite of the fact that investee firms' (negative) inside-out impact can profoundly, and adversely, affect the economy at large, and thereby pose financially material risks to the portfolio.¹⁵⁸ That view is evidently shared by the European Commission, whose CSRD Proposal relies, amongst other things, on the consideration that the two perspectives of the double-materiality principle for nonfinancial reporting are 'often not well understood or applied', making it necessary to both 'clarify that undertakings should consider each materiality perspective in its own right, and should disclose information that is material from both perspectives as well as information that is material from only one perspective', 159 and compel reporting entities to embrace EU-mandated double-materiality principled common reporting

¹⁵⁵ ShareAction (2021), p 27.

¹⁵⁶ Ibid., pp 29-30.

¹⁵⁷ Ibid., p 28.

¹⁵⁸ See ShareAction (2020), pp 5, 19-21.

¹⁵⁹ Recital 25 to the CSRD Proposal, as amended by Coreper.

standards to ensure that all relevant information that is material to users is disclosed.160

Nevertheless, the fact remains that low rates of voting support from the largest asset managers to environmental and social proposals seem to be in contrast with the likely preferences of E&S-minded end-investors. 161

One possible way by which to align asset managers' E&S votes with the actual preferences of end-investors is to overcome the judgement of stewardship teams, and the associated biases, by resorting to pass-through voting or end-client voting instructions based on the format 'If X, do Y', i.e., mechanisms by which to ultimately transfer the power to decide how to vote the shares in the fund from the fund to their investors. 162 As the fund managers would be required to follow end-investor-generated guidelines when voting the shares, such mechanisms would force funds to split the exercise of voting rights proportionately rather than voting as an undifferentiated bloc; which would remove some of the largest asset managers' discretionary power and reduce their concentrated voting influence.¹⁶³ While, famously, pass-through voting was actually launched by BlackRock as a first-mover in January 2022 for institutional clients invested in certain pooled vehicles managed in the US and UK,164 some (equally large) asset managers may perhaps follow the lead in the future. After all, pass-through voting might also serve as a lever for offering competitive advantage to a firm's current and potential clients.

As pass-through voting is not possible in any jurisdiction from a regulatory standpoint, regulators, too, may consider assessing whether to allow end-investors who wish so to access similar tools on a legal basis.

¹⁶⁰ Recital 32 to the CSRD Proposal, as amended by Coreper. See also European Commission (2021c), pp 57-58.

¹⁶¹ See Griffin (2020), pp 439-440.

¹⁶² Ibid., p 440.

¹⁶³ Ibid., p 441. See also Griffin (2020b), p 964.

¹⁶⁴ On 13 June 2022, BlackRock announced it would be increasing the range of funds eligible for its 'Voting Choice' program in the UK and expand the program to Canadian and Irish pooled funds. Nearly half (47%) of the \$4.9 trillion index equity assets are now eligible to participate in the firm's program. Following the changes announced, voting choice is available for 100% of US pension plans, and 95% of the firm's institutional index equity funds (amounting to about half of the firm's index equity assets and virtually all of its index equity assets outside ETFs and retail mutual funds). In Europe and the UK, 80% of BlackRock's index equity assets (other than ETFs) are eligible for the program. See BlackRock (2022a). Blackrock also published a white paper outlining the firm's ambition to expand Voting Choice to all investors, including individual investors in funds: see Blackrock (2022b). According to the firm, BlackRock clients have committed \$530 billion or a quarter of eligible assets -to voting their own preferences through Voting Choice. Of these, clients representing \$120 billion of assets have elected to vote their own preferences in the five months since BlackRock introduced the program: ibid., p 4.

Accurately accounting for the many challenges and unknowns such a move indeed poses would however be imperative. 165

To begin with, pass-through voting is likely, at present, to be most impractical for the vast majority of not-so-large asset managers, let alone considering its costliness and operational complexity. ¹⁶⁶ But, in turn, if pass-through voting were adopted by larger asset managers only, this might further entrench their competitive power over smaller players. Further, 'passing voting authority for hundreds of companies to investors would not only be overwhelming for the fund, but also for investors', unless some corrective mechanism be adopted, such as restricting the rule to specified 'non-routine' matters, amongst which E&S issues. ¹⁶⁷

Moreover, based on experience with early adopters of BlackRock's voting choice program, it is not unlikely that many institutional end-clients would eventually end up instructing the asset manager to vote according to some partnering proxy advisor's voting policy selected out of a given menu.¹⁶⁸ Even if end-client choice would probably achieve a better alignment of the votes cast with its own preferences (as mirrored in the proxy advisor policy chosen), still this kind of pass-through voting is a way of shifting voting authority from one agent (the asset manager) to another (the proxy advisor) who is not free from criticism either. Were pass-through voting to be extended to individual shareholders in the funds, proxy advisor-based voting would probably be the outcome of most voting-choice adopters. The issues of rational apathy, information cost, and financial illiteracy, which still prevent large cohorts of retail stockholders from voting their shares, would predictably prevent individual retail shareholders in the fund from choosing to instruct the asset manager based on any option requiring them to directly make a voting decision - except, perhaps, for younger, and technology-at-ease, investor generations that claim for their voice to be heard, particularly where ESG issues are at stake at portfolio companies. Finally, pass-through voting can make it very hard for portfolio firms to predict shareholders' voting behavior, adding further challenges on the board when defining corporate E&S strategy.

7 Conclusions

The strong regulatory signals on sustainable finance ignited by the implementation of the EU's Sustainable Finance Agenda, which set clear

¹⁶⁵ See, e.g., Lipton (2022).

¹⁶⁶ See Stevens (2018).

¹⁶⁷ Lund (2018), p 530.

¹⁶⁸ See BlackRock (2022b), p 10.

priorities and expectations of investors, may well be among the factors contributing to European asset managers' overall stronger performance concerning their approach to responsible investing and engaging relative to US managers and the rest of the world. It may also be the case that investors' approach to sustainable investing and ESG-related stewardship will improve over time and become more effective. This however requires, at the very least, that more reliable, in terms of heightened transparency, granularity and comparability of ESG ratings and indices be made available, and be made available at a fair price. Where no such condition is granted, key determinants of asset managers' decisions 'can be over simplistic and may not return the desired results'. The same is true with respect to corporate non-financial disclosures, where international convergence of disclosure standards is warranted too.

The regulatory efforts in the EU clearly highlight the fact that, in an ever globalizing market for investing, in which investments are typically crossborder, action is needed on a much larger scale than just at the level of the EU in order for sustainability objectives to be more credibly met, or even only neared.¹⁷¹ Regulatory initiatives regional in scope cannot, almost by definition, address challenges that are global in nature. It is true that the scope of application of EU legislation on sustainable finance is indeed very broad. For example, disclosure obligations imposed by the Taxonomy Regulation also cover third-country financial market participants that sell financial products in the EU, and companies subject to the NFRD regardless of their location.¹⁷² Non-EU investors offering products in Europe are subject to the SFDR as well. The extra-territorial reach of EU regulations in the field of sustainable finance may perhaps support a 'Brussels effect',173 to some extent, since investors and companies might choose to extend the EU requirements to their global operations, if the benefits of standardization around a specified regulatory standard outweigh the costs of differently managing operations in the EU and the rest of the world.

Yet, limitations on the role to be possibly played by institutional investors in the transition towards sustainable growth are unlikely to be easily overcome. First, the expectation that institutions invest resources with the aim of promoting socially responsible conduct and achieving particular ESG objectives entails an internalization of costs by institutional

¹⁶⁹ See ShareAction (2020), p 27; Lafarre (2022), p 31.

¹⁷⁰ See Boffo and Patalano (2020), p 36.

¹⁷¹ See generally Busch (2021).

 $^{^{172}}$ See European Commission and Technical Expert Group on Sustainable Finance (2020), p 10.

¹⁷³ Bradford (2020); Bradford (2012), p 3. The Brussels Effect refers to how EU rules and regulations can penetrate outside the EU through market mechanisms, resulting in the globalization of standards.

investors,¹⁷⁴ the effects of which will be felt not only by them (in terms of the lower net earnings) but also by end investors (in terms of the lower returns). Therefore, even assuming that largely diversified institutional investors adopt a portfolio value maximization approach according to which they can push ESG policies that can impair the value of some portfolio companies while benefiting some others,¹⁷⁵ the fact remains that '[a] rational owner would use his power to internalize externalities so long as its share of the cost to the externality-causing firms are lower than the benefits that accrue to the entire portfolio from the elimination of the externality'.¹⁷⁶ Coupled with further disincentives, as described above, this reasoning justifies scepticism as to whether institutional investors can perform a 'public' function for the benefit of society at large and restrain the need for governmental or sharper regulatory intervention.

References

- Azar J, Duro M, Kadach I, Ormazabal G (2021) The Big Three and corporate carbon emissions around the world, J. Fin. Econ. 142(2) 674-696. https://doi.org/10.1016/j.jfineco.2021.05.007
- Barzuza M, Curtis Q, Webber DH (2020) Shareholder value(s): index fund ESG activism and the new millennial corporate governance. S Cal L Rev 93(6):1243-1321
- Bebchuk L, Hirst S (2019) Index funds and the future of corporate governance: theory, evidence, and policy. Colum L Rev 119(8):2029-2146
- Bebchuk L, Hirst S (2022) The power of the Big Three and why it matters. B U L Rev 2022 (forthcoming). http://www.law.harvard.edu/faculty/bebchuk/The_Power_of_the_Big_Three_and_Why_It_Matters.pdf. Accessed 13 September 2022
- Becht M, Franks J, Mayer C, Rossi S (2010) Returns to shareholder activism: evidence from a clinical study of the Hermes UK Focus Fund. Rev Fin Stud 23(3):3093-3129. https://doi.org/10.1093/rfs/hhn054.ra
- Bhagat D (2022) An inconvenient truth about ESG investing. Harv Bus Rev (Digital Articles), 31 March 2022. https://hbr.org/2022/03/an-inconvenient-truth-about-esg-
 - investing#:~:text=Unfortunately%20ESG%20funds%20don't,in%202%2C428%20non%2DESG%20portfolios. Accessed 13 September 2022
- Bioy H (2017) Passive fund providers take an active approach to investment stewardship. Morningstar, 6 December 2017. https://www.morningstar.com/articles/839337/passive-fund-providers-take-an-active-approach-to-investment-stewardship. Accessed 13 September 2022

-

¹⁷⁴ See generally Johnston et al. (2021), p 39.

¹⁷⁵ See Enriques (2020).

¹⁷⁶ Condon (2020), p 6.

- BlackRock Investment Institute (2020) Sustainability: the tectonic shift transforming investing. https://www.blackrock.com/us/individual/insights/blackrock-investment-institute/sustainability-in-portfolio-construction. Accessed 13 September 2022
- BlackRock (2020) BlackRock investment stewardship. https://www.blackrock.com/corporate/literature/publication/blk-profile-of-blackrock-investment-stewardship-team-work.pdf. Accessed 13 September 2022
- BlackRock (2022a) BlackRock expands voting choice to additional clients. Press release, 13 June 2022. https://www.blackrock.com/corporate/newsroom/press-releases/article/corporate-one/press-releases/2022-blackrock-voting-choice.
 Accessed 13 September 2022
- Blackrock (2022b) It's all about choice. BISM0622U/M-2240317-3/21. https://www.blackrock.com/corporate/literature/publication/its-all-about-choice.pdf. Accessed 13 September 2022
- Boffo R, Patalano R (2020) ESG Investing: practices, progress and challenges. OECD, Paris. https://www.oecd.org/finance/ESG-Investing-Practices-Progress-Challenges.pdf. Accessed 13 September 2022
- Bolton B, Kacperczyk M (2021) Do investors care about carbon risk? J Fin Econ 142(2):517-549. https://doi.org/10.1016/j.jfineco.2021.05.008
- Bonacchi N, Klein A, Longo S, Strampelli G (2022) The effects of credible voluntary disclosures: institutional investor engagement and investees' ESG performances. European Corporate Governance Institute Law Working Paper No. 622/2022. http://dx.doi.org/10.2139/ssrn.4011957. Accessed 13 September 2022
- Bradford A (2020) The Brussels Effect: how the European Union rules the world. OUP, New York. https://doi.org/10.1093/oso/9780190088583.001.0001
- Bradford A (2012) The Brussels Effect. NW U L Rev 107(1):1-68.
- Busch D (2021) Sustainability disclosure in the EU financial sector. European Banking Institute Working Paper Series 2021, n. 70. http://dx.doi.org/10.2139/ssrn.3650407. Accessed 13 September 2022
- Business Roundtable (2019) Business Roundtable redefines the purpose of a corporation to promote 'an economy that serves all Americans. 19 August 2019. https://www.businessroundtable.org/business-roundtable-redefines-the-purpose-of-a-corporation-to-promote-an-economy-that-serves-all-americans. Accessed 13 September 2022
- Carleton WT, Nelson JM, Weisbach MS (1998) The influence of institutions on corporate governance through private negotiations: evidence from TIAA-CREF. J Fin 53(4):1335-1362. https://doi.org/10.1111/0022-1082.00055
- Chalmers J, Cox E, Picard N (PwC) (2021) The economic realities of ESG. PwC, 28 October 2021. https://www.pwc.com/economic-realities-of-ESG. Accessed 13 September 2022.
- Chiu I H-Y (2022) The EU sustainable finance agenda: developing governance for double materiality in sustainability metrics. EBOR 23:87-123. https://doi.org/10.1007/s40804-021-00229-9
- Coates JC, IV, Glenn Hubbard G (2007) Competition in the mutual fund industry: evidence and implications for policy. J Corp L 33(1):151-222
- Condon M (2020) Externalities and the common owner. Wash L Rev 95(1):1-81

- Corporate Governance Committee (2019) The 2020 Belgian Code on Corporate Governance. https://www.corporategovernancecommittee.be/en/over-de-code-2020/2020-belgian-code-corporate-governance. Accessed 13 September 2022
- Davies P (2020) The UK Stewardship Code 2010-2020. From saving the company to saving the planet? European Corporate Governance Institute Law Working Paper No. 506/2020. http://dx.doi.org/10.2139/ssrn.3553493. Accessed 13 September 2022
- De La Cruz A, Medina A, Tang Y (2019) Owners of the world's listed companies. OECD. http://www.oecd.org/corporate/Owners-of-the-Worlds-Listed-Companies.pdf. Accessed 13 September 2022
- Demers E, Hendrikse J, Joos P, Lev B (2021) ESG did not immunize stocks during the COVID-19 crisis, but investments in intangible assets did. J Bus Fin Acc 2021(48):433-462. https://doi.org/10.1111/jbfa.12523
- Enriques L (2020) Missing in today's shareholder value maximization credo: the shareholders. ProMarket, 22 September 2020. https://promarket.org/2020/09/22/milton-friedman-value-maximization-credo-is-missing-the-shareholders/. Accessed 13 September 2022
- European Commission (2012) Action Plan: European company law and corporate governance. A modern legal framework for more engaged shareholders and sustainable companies, 12 December 2012, COM(2012) 740 final. https://eurlex.europa.eu/LexUriServ/LexUriServ.do?uri=COM:2012:0740:FIN:EN:PDF. Accessed 13 September 2022
- European Commission (2014) Proposal for a Directive of the European Parliament and of the Council Amending Directive 2007/36/EC as regards the encouragement of long-term shareholder engagement and Directive 2013/34/EU as regards certain elements of the corporate governance statement, 9 April 2014, COM(2014) 213 final, 2014/0121 (COD). https://eurlex.europa.eu/resource.html?uri=cellar:59fccf6c-c094-11e3-86f9-01aa75ed71a1.0003.01/DOC_1&format=PDF. Accessed 13 September 2022
- European Commission (2018a) Communication from the Commission to the European Parliament, the European Council, the Council, the European Central Bank, the European Economic and Social Committee and the Committee of the Regions, Action Plan: Financing Sustainable Growth, COM(2018) 97 final. https://eurlex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:52018DC0097&from=EN. Accessed 13 September 2022
- European Commission (2018b) Renewed sustainable finance strategy and implementation of the action plan on financing sustainable growth, 8 March 2018. https://ec.europa.eu/info/publications/sustainable-finance-renewed-strategy_en. Accessed 13 September 2022
- European Commission (2019) The European Green Deal. Communication from the Commission to the European Parliament, the European Council, the Council, the European Economic and Social Committee and the Committee of the Regions, 11 December 2019, COM(2019) 640 final. https://eurlex.europa.eu/resource.html?uri=cellar:b828d165-1c22-11ea-8c1f-01aa75ed71a1.0002.02/DOC_1&format=PDF. Accessed 13 September 2022
- European Commission (2020) Sustainable Corporate Governance Initiative. https://ec.europa.eu/info/law/better-regulation/have-your-

- <u>say/initiatives/12548-Sustainable-corporate-governance_en</u>. Accessed 13 September 2022
- European Commission (2021a) Strategy for financing the transition to a sustainable economy. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions, 6 July 2021, COM(2021) 390 final. https://eurlex.europa.eu/resource.html?uri=cellar:9f5e7e95-df06-11eb-895a-01aa75ed71a1.0001.02/DOC_1&format=PDF. Accessed 13 September 2022
- European Commission (2021b) Proposal for a Directive of the European Parliament and of the Council amending Directive 2013/34/EU, Directive 2004/109/EC, Directive 2006/43/EC and Regulation (EU) No 537/2014, as regards corporate sustainability reporting, 21 April 2021, COM(2021) 189 final, 2021/0104 (COD). https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:52021PC0189&from=EN. Accessed 13 September 2022
- European Commission (2021c) Report from the Commission to the European Parliament, the Council and the European Economic and Social Committee on the review clauses in Directives 2013/34/EU, 2014/95/EU, and 2013/50/EU, 21 April 2021, COM(2021) 199 final. https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:52021DC0199&from=EN. Accessed 13 September 2022
- European Commission (2021d) Commission Staff Working Document. Fitness Check on the EU framework for public reporting by companies, 21 April 2021, SWD(2021) 81 final. https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:52021SC0081&from=EN. Accessed 13 September 2022
- European Commission (2022a) Proposal for a Directive of the European Parliament and of the Council on Corporate Sustainability Due Diligence and amending Directive (EU) 2019/1937, 23 February 2022, COM(2022) 71 final, 2022/0051 (COD). https://eur-lex.europa.eu/resource.html?uri=cellar:bc4dcea4-9584-11ec-b4e4-01aa75ed71a1.0001.02/DOC_1&format=PDF. Accessed 13 September 2022
- European Commission (2022b) Summary Report. Targeted consultation on the functioning of the ESG ratings market in the EU and on the consideration of ESG factors in credit ratings. https://ec.europa.eu/info/sites/default/files/business_economy_euro/banking_and_finance/documents/2022-esg-ratings-summary-of-responses_en.pdf. Accessed 13 September 2022
- European Commission and Technical Expert Group on Sustainable Finance (2020)
 Frequently Asked Questions about the work of the European Commission and the Technical Expert Group on Sustainable Finance on EU Taxonomy & Green Bond Standard, 10 June 2020, p. 10.

 https://ec.europa.eu/info/sites/default/files/business_economy_euro/banking_and_finance/documents/200610-sustainable-finance-teg-taxonomy-green-bond-standard-faq_en.pdf. Accessed 13 September 2022
- European Court of Justice (2009) Case C-45/08, Spector Photo Group NV and Chris Van Raemdonck v Commissie voor het Bank-, Financie- en Assurantiewezen (CBFA) [2009]. Judgment of the Court (Third Chamber), 23 December 2009. ECR I-12073. https://eur-lex.europa.eu/legal-

- <u>content/EN/TXT/PDF/?uri=CELEX:62008CJ0045&from=EN</u>. Accessed 13 September 2022
- European Fund and Asset Management Association (2018) Stewardship Code 2018. https://www.efama.org/Publications/Public/Corporate_Governance/EFA MA%20Stewardship%20Code.pdf. Accessed 13 September 2022
- European Fund and Asset Management Association (2021) Asset management in Europe. An overview of the asset management industry, 13th edn, facts and figures.

 December

 2021.

 https://www.efama.org/sites/default/files/files/Asset%20Management%2

 0Report%202021_3.pdf. Accessed 13 September 2022
- European Fund and Asset Management Association (2022a) Response to the European Commission consultation on the functioning of the ESG ratings market in the EU and on the consideration of ESG factors in credit ratings, 8 June 2022. https://www.efama.org/newsroom/news/efama-response-ec-cfe-esg-ratings-and-sustainability-risks-credit-ratings. Accessed 13 September 2022
- European Fund and Asset Management Association (2022b) Fact Book 2022. https://www.efama.org/sites/default/files/files/Fact%20Book%202022%20 lowres_2.pdf. Accessed 13 September 2022
- European Securities and Markets Authority (2019) Report on undue short-term pressure on corporations, 18 December 2019, ESMA30-22-762. https://www.esma.europa.eu/document/report-undue-short-term-pressure-corporations-financial-sector. Accessed 13 September 2022
- European Securities and Markets Authority (2021) Letter to the European Commission, 28 January 2021, ESMA30-379-423. https://www.esma.europa.eu/sites/default/files/library/esma30-379-423_esma_letter_to_ec_on_esg_ratings.pdf. Accessed 13 September 2022
- European Securities and Markets Authority (2022) Performance and costs of EU retail investment products. ESMA Annual Statistical Report 2022, 5 April 2022, ESMA 50-165-1677. https://www.esma.europa.eu/sites/default/files/library/esma_50-165-1677_asr_performance_and_costs_of_eu_retail_investment_products.pdf. Accessed 13 September 2022
- Fairfax LM (2013) Mandating board-shareholder engagement? U Ill L Rev 2013:821-857
- Financial Reporting Council (2018) UK Corporate Governance Code 2018. https://www.frc.org.uk/directors/corporate-governance-and-stewardship/uk-corporate-governance-code. Accessed 13 September 2022
- Financial Reporting Council (2019) UK Stewardship Code 2020. https://www.frc.org.uk/getattachment/5aae591d-d9d3-4cf4-814a-d14e156a1d87/Stewardship-Code_Dec-19-Final-Corrected.pdf. Accessed 13 September 2022
- Fisch JE, Hamdani A, Davidoff Solomon S (2019) The new titans of Wall Street: a theoretical framework for passive investors, U Pa L Rev 168:17-72
- Fink L (2021) Larry Fink's 2021 letter to CEOs. https://www.blackrock.com/us/individual/2021-larry-fink-ceo-letter. Accessed 13 September 2022
- Fink L (2022) The power of capitalism. Larry Fink's 2022 letter to CEOs. https://www.blackrock.com/corporate/investor-relations/larry-fink-ceo-letter. Accessed 13 September 2022

- Fleischer H (2009) Investor Relations und informationelle Gleichbehandlung im Aktien-, Konzern- und Kapitalmarktrecht. Zeitschrift für Unternehmens- und Gesellschaftsrecht 38(4):505-541. https://doi.org/10.1515/ZGRE.2009.505
- Franks J (2020) Institutional ownership and governance. Oxford Rev Ec Pol 36(2):258-274. https://doi.org/10.1093/oxrep/graa009
- Friedman M (1970) The social responsibility of business is to increase its profits. The New York Times Magazine, 13 September 1970
- Global Sustainable Investment Alliance (2021) Global Sustainable Investment Review 2020. http://www.gsi-alliance.org/wp-content/uploads/2021/08/GSIR-20201.pdf. Accessed 13 September 2022
- Goldman Sachs Equity Research (2019) Shareholder engagement in the age of transparency, 12 June 2019. https://www.eticanews.it/wp-content/uploads/2019/06/Shareholder-engagement-in-the-age-of-transparency-1.pdf. Accessed 13 September 2022
- Gordon JN (2022) Systematic stewardship, 24 January 2022. J Corp L (forthcoming), European Corporate Governance Institute Law Working Paper No. 566/2021, Columbia Law and Economics Working Paper No. 640. http://dx.doi.org/10.2139/ssrn.3782814. Accessed 13 September 2022
- Griffin CN (2020) Margins: estimating the influence of the Big Three on shareholder proposals. SMU L Rev 73:409-443
- Griffin CN (2020b) We three kings: disintermediated voting at the index fund giants? Md L Rev 79(4):954-1008
- Hartzmark SM, Sussmann AM (2019) Do investors value sustainability? A natural experiment examining ranking and fund flows. J Fin 74(6): 2789-2837. https://doi.org/10.1111/jofi.12841
- Hirt H-C, Hopt KJ, Mattheus D (2016) Dialog zwischen dem Aufsichtsrat und Investoren, Rechtsvergleichende und rechtsdogmatische Überlegungen zur Investorenkommunikation in Deutschland. AG 2016(20): 725-738
- Hopt KJ (2011) Comparative corporate governance: the state of the art and international regulation. Am J Comp L 59(1):1-73. https://doi.org/10.5131/AJCL.2010.0025
- International Organization of Securities Commissions (2021) Environmental, Social and Governance (ESG) Ratings and Data Products Providers Final Report, November 2021, FR09/21. https://www.iosco.org/library/pubdocs/pdf/IOSCOPD690.pdf. Accessed 13 September 2022
- Italian Corporate Governance Committee (2020) Corporate Governance Code. https://www.borsaitaliana.it/comitato-corporate-governance/codice/2020eng.en.pdf. Accessed 13 September 2022
- Johnston A, Amaeshi K, Adegbite E, Osuji O (2021) Corporate social responsibility as obligated internalisation of social costs. J Bus Ethics 170:39-52. https://link.springer.com/article/10.1007/s10551-019-04329-y
- Kahan M, Rock E (2020) Index funds and corporate governance: let shareholders be shareholders. B U L Rev 100(5):1771-1815
- Kelly TG (2021) Institutional investors as environmental activists. J Corp L Stud 21(2):467-489. https://doi.org/10.1080/14735970.2021.1881356
- Lafarre A (2022) Do institutional investors vote responsibly? TILEC Discussion Paper No. DP2022-001, Tilburg Law School Research Paper, 16 February 2022. http://dx.doi.org/10.2139/ssrn.4042907. Accessed 13 September 2022

- Lazard (2022) 2021 Review of Shareholder Activism, 19 January 2022. https://www.lazard.com/media/452017/lazards-2021-review-of-shareholder-activism_vff.pdf. Accessed 13 September 2022
- Lazard (2021) 2020 Review of Shareholder Activism, 21 January 2021. https://www.lazard.com/media/451536/lazard s-2020-review-of-shareholder-activism-vf.pdf. Accessed 13 September 2022
- Lipton A (2022) Power concedes nothing without a demand. Bus Law Prof Blog, 5
 March
 2022.
 https://lawprofessors.typepad.com/business_law/2022/03/power-concedes-nothing-without-a-demand.html. Accessed 13 September 2022
- Lombardo S, Mucciarelli FM (2017) Market soundings: the interaction between securities regulation and company law in the United Kingdom and Italy. European Corporate Governance Institute Law Working Paper No. 362/2017. https://ssrn.com/abstract=3012183. Accessed 13 September 2022
- Lopez deSilanes F, McCahery JA, Pudschedl PC (2019) ESG performance and disclosure: a cross-country analysis. TILEC Discussion Paper No. DP2019-032, European Corporate Governance Institute Law Working Paper No. 418/2019. http://dx.doi.org/10.2139/ssrn.3506084. Accessed 13 September 2022
- Lopez de Silanes F, McCahery JA, Pudschedl PC (2022) Institutional investors and ESG preferences. European Corporate Governance Institute Law Working Paper No. 631/2022. http://dx.doi.org/10.2139/ssrn.4049313. Accessed 13 September 2022
- Lund DS (2018) The case against passive shareholder voting. J Corp L 43(3):493-536 Lund DS (2022) Asset managers as regulators. U Pa L Rev 171, 2022 (forthcoming), USC CLASS Research Paper No. CLASS22-12, USC Law Legal Studies Paper No. 22-12. http://dx.doi.org/10.2139/ssrn.3975847. Accessed 13 September 2022
- Mahoney PG, Robertson AZ (2021) Advisers by another name. Harv Bus L Rev 11(2):311-358
- McCahery JA, Sautner Z, Starks LT (2016) Behind the scenes: the corporate governance preferences of institutional investors. J Fin 71(6):2905-2932
- Moisson E (2021) Net inflows into UBS ETFs this year have been only to ESG funds. The Financial Times, 17 November 2021. https://www.ft.com/content/c9f8e362-edb7-4693-92da-a57e77652850. Accessed 13 September 2022
- Mooney A (2021) Stay or sell? The \$110tn investment industry gets tougher on climate. The Financial Times, 15 November 2021. https://www.ft.com/content/ee08d61d-4c98-4398-9971-93036d67e91e. Accessed 13 September 2022
- Morningstar (2021) Global Sustainable Fund Flows: Q3 2021 in Review, November 2021. https://www.morningstar.com/lp/global-esg-flows. Accessed 13 September 2022.
- Pacces AM (2021) Will the EU Taxonomy Regulation foster sustainable corporate governance? Sustainability 13(21), 12316. https://doi.org/10.3390/su132112316
- Posner C (2021) BlackRock to permit some clients to vote. Harvard L Sch Forum on Corporate Governance, 25 October 2021. https://corpgov.law.harvard.edu/2021/10/25/blackrock-to-permit-some-clients-to-vote/. Accessed 13 September 2022

- PricewaterhouseCoopers (2021) PwC's 2021 Global Investor Survey. December 2021. https://www.pwc.com/gx/en/services/audit-assurance/corporate-reporting/2021-esg-investor-survey.html. Accessed 13 September 2022
- Quinio A (2022) Morningstar cuts 1,200 funds from 'sustainable' list. The Financial Times, 10 February 2022. https://www.ft.com/content/9cf8c788-6cad-4737-bc2a-2e85ac4aef7d. Accessed 13 September 2022
- Raghunandan A, Rajgopal S (2022) Do ESG funds make stakeholder-friendly investments? Rev Acc't Stud 27:822-863. https://doi.org/10.1007/s11142-022-09693-1
- Ringe W-G (2021a) Investor-led Sustainability in corporate governance. European Corporate Governance Institute Law Working Paper No. 615/2021, LawFin Working Paper No. 36, Annals of Corporate Governance, 2022 (forthcoming). http://dx.doi.org/10.2139/ssrn.3958960. Accessed 13 September 2022
- Ringe W-G (2021b) Stewardship and shareholder engagement in Germany. EBOR 22:87-124. https://doi.org/10.1007/s40804-020-00195-8
- Robertson AZ (2019) Passive in name only: delegated management and index investing. Yale J on Reg 36(2):795-852
- Rosati N, Bomprezzi P, Ferraresi M, Frigo A, Nardo M (2020) Common shareholding in Europe. EUR 30312 EN, Publications Office of the European Union, Luxembourg, 2020, https://publications.jrc.ec.europa.eu/repository/handle/JRC121476. doi:10.2760/734264. Accessed 13 September 2022
- ShareAction (2020) Point of no returns. A ranking of 75 of the world's largest asset managers' approaches to responsible investment. March 2020. https://api.shareaction.org/resources/reports/Point-of-no-Returns.pdf. Accessed 13 September 2022
- ShareAction (2021) Voting Matters 2021. Are asset managers using their proxy votes for action on environmental and social issues? December 2021. https://shareaction.org/reports/voting-matters-2021-are-asset-managers-using-their-proxy-votes-for-action-on-environmental-and-social-issues. Accessed 13 September 2022
- Sharfman BS (2022) Looking at the 'Big Three' investment advisers through the lens of agency. Oxford Business Law Blog, 18 February 2022. https://www.law.ox.ac.uk/business-law-blog/blog/2022/02/looking-big-three-investment-advisers-through-lens-agency. Accessed 13 September 2022
- Sharfman BS (2022b) Opportunism in the shareholder voting and engagement of the 'Big Three' investment advisers to index funds. 5 February 2022, 48 J Corp L (forthcoming). http://dx.doi.org/10.2139/ssrn.3995714. Accessed 13 September 2022
- Soltes E (2018) What can managers privately disclose to investors? Yale J Reg Bulletin 36: 148-169
- Somerset Webb M (2022) Divesting fossil fuel stocks? That's so last year. The Financial Times, 18 February 2022. https://www.ft.com/content/b159c919-ddef-420c-b63b-f90441e8c758. Accessed 13 September 2022
- Standard Chartered Bank (2021) Sustainable Investing Review 2021. https://www.sc.com/en/banking/banking-for-individuals/wealth-management/standard-chartered-sustainable-investing-review-2021/.

 Accessed 13 September 2022

- Steuer S (2022) Common ownership and the (non-)transparency of institutional shareholdings: an EU-US comparison, 25 July 2022, SAFE Working Paper No. 354. http://dx.doi.org/10.2139/ssrn.4171508. Accessed 13 September 2022
- Stevens PS (2018) SEC should reject complex, costly 'pass-through' proxy voting. Investment Company Institute, 2 October 2018. https://www.ici.org/viewpoints/view-18-passthrough-voting. Accessed 13 September 2022
- Strampelli G (2021) Can BlackRock save the planet? The institutional investors' role in stakeholder capitalism. Harv Bus L Rev 11:1-20
- Strine LE, Jr (2020) Stewardship 2021: the centrality of institutional investor regulation to restoring a fair and sustainable American economy. U Pa Inst for Law & Econ Research Paper No. 20-55, 2020; Columbia Law and Economics Working Paper No. 633. http://dx.doi.org/10.2139/ssrn.3719145. Accessed 13 September 2022
- Tett G, Nauman B, Temple-West P, Edgecliffe-Johnson A (2020) ESG shines in the crash; legal milestone for ratings. The Financial Times, 13 March 2020. https://www.ft.com/content/dd47aae8-ce25-43ea-8352-814ca44174e3. Accessed 13 September 2022
- Thomas RS, Palmiter AR, Cotter JF (2012) Dodd-Frank's say on pay: will it lead to a greater role for shareholders in corporate governance. Cornell L Rev 97(5):1213-1266
- World Economic Forum (2019) Davos Manifesto 2020: the universal purpose of a company in the Fourth Industrial Revolution, 2 December 2019. https://www.weforum.org/agenda/2019/12/davos-manifesto-2020-the-universal-purpose-of-a-company-in-the-fourth-industrial-revolution. Accessed 13 September 2022
- Zetzsche DA, Anker-Sørensen L (2022) Regulating sustainable finance in the dark. EBOR 23:47-85. https://doi.org/10.1007/s40804-021-00237-9